CST Part IA Regular Languages and Finite Automata List of corrections to the lecture notes May 4, 1999

Page 19, Slide 18: Replace Şδ_{P M} (S)Ť by Şδ_{P M} (S, a)Ť, twice.

Page 19: In the line labelled (1), delete the second occurrence of Şq_1Ť (or replace it by Şq_2Ť). Page 19, line -1: Şwhere S_1 = δ_{P M} (s_{P M} , a_1), S_2 = δ_{P M} (S_1, a_2)Ť. Page 20, line 2: Şδ_{P M} (s_{P M} , a_1)Ť. Page 20, line 3: Şδ_{P M} (S_1, a_2)Ť.

Page 20, lines 5 and 13: Şδ_{P M} (S_n�_1, a_n)Ť. Page 20, line 14: Şq_n�_1^an) q_nŤ. Page 20, line 14: Şδ_{P M} (S_n�_2, a_n�_1)Ť. Page 20, line 15: Şq_n�_2^an�^1) q_n�_1Ť.

Page 51, lines -2 and -4: Replace Şs_mŤ by Şs_M Ť (twice).